Title: Values Clarification in a decision aid about fertility preservation: does it add to information provision? Two randomized experiments with healthy participants

Authors:

Mirjam M Garvelink (m.m.garvelink@lumc.nl)
Moniek M ter Kuile (m.m.ter_kuile@lumc.nl)
Anne M Stiggelbout (a.m.stiggelbout@lumc.nl)
Marieke de Vries (mariekedevries@uvt.nl)

Version: 2
Date: 4 January 2014

Author's response to reviews: see over
Dear Dr Wyatt,

As was requested in a pre-review request, the following changes were made to the manuscript “Values Clarification in a decision aid about fertility preservation: does it add to information provision? Two randomized experiments with healthy participants (MS: 1848265293117094)” in order to finalize the manuscript for the review process:

1. More author details were added to the title page of the manuscript (i.e. email addresses of all authors).
2. For both experiments more details were added about ethics (that the experiments were performed in accordance with the declaration of Helsinki, and if and where the studies were approved by ethical committees), and that participants all gave their informed consent before participating.
3. The sentence declaring no conflicts of interest was moved from the acknowledgement section to a separate section with subheading "conflicts of interest".

I hope the manuscript in its current form is ready to send out for peer review.

Yours sincerely,

Mirjam Garvelink